Author's response to reviews

Title: A Multi-center, Randomized, Clinical Study to Compare the Effect and Safety of Autologous Cultured Osteoblast Injection to Treat Fractures

Authors:

Seok-Jung Kim M.D. (peter@catholic.ac.kr)
Yong-Woon Shin M.D. (woonysos@daum.net)
Kyu-Hyun Yang M.D. (kyang@yumc.yonsei.ac.kr)
Sang-Bum Kim M.D. (sangbume@hitel.net)
Moon-Jib Yoo M.D. (miyoo@dku.edu)
Suk-Ku Han M.D. (hnsukku@catholic.ac.kr)
Soo-Ah Im M.D. (saim@catholic.ac.kr)
Yoo-Dong Won M.D. (yoodong1@catholic.ac.kr)
Yerl-Bo Sung M.D. (ybs58@sanggyepaik.ac.kr)
Taek-Soo Jeon M.D. (sirjeon@kyuh.co.kr)
Cheong-Ho Chang M.D. (chc@swcell.com)
Jae-Deog Jang Ph.D. (idjang@swcell.com)
Hyun-Jo Kim M.S. (angel@swcell.com)
Su-Young Lee M.S. (sushine@swcell.com)

Version: 7 Date: 24 October 2008

Author's response to reviews: see over
Reviewer's report
Title: A Multi-center, Randomized, Clinical Study to Compare the Effect and Safety of Autologous Cultured Osteoblast Injection to Treat Fractures
Version: 5 Date: 2 October 2008
Reviewer: Jennifer Westendorf
Reviewer's report:
Kim and colleagues have submitted a revised manuscript that describes the results of a multi-center randomized clinical trial that compared the safety and efficacy of autologously cultured osteoblasts on fracture treatment. In the revised manuscript, the authors addressed all the previous concerns; however, they were only partially successful in answering them adequately. The lingering compulsory issues are:

1. Figure 4 is a new figure. Are the data from each time point statistically significant between the control and treated groups?

   ➔ After two months, the data were statistically significant. But at one month, the data were not statistically significant. I inserted this fact in the result section.

2. The fracture healing data was separated into age groups as requested and added as Figure 6. The authors state that there was no statistical difference between the two groups. Which two groups? How do the 20-29 year olds compare to each of the other groups? They seemed to respond the best.

   ➔ There was no statistical difference in the osteoblast injection response between the younger age patient group and the older age group (p=0.71); also, when we compared the 20-29-year-old patient group with other groups, there was no statistical difference. (Fig. 5)

3. The control group was untreated and did not receive fibrin injections. On page 12, the authors provide a nice discussion of the potentially beneficial effects of fibrin on growth factor recruitment and retention. Given these potentially positive effects, the authors should add a sentence stating that they cannot conclude that the fibrin did not have an effect in this study.

   ➔ As perhaps the sentence on page 12 caused a misunderstanding about fibrin, we changed it. We intended to communicate that fibrin just acts as a carrier when we attempt to apply a cell or growth factor to a certain area as fibrin does not induce growth factor recruitment. Injection only fibrin to fracture site is ethically problematic. The injection procedure involves some stress, time, and patient inconvenience, and no patient will agree a treatment that is ineffective for them.
4. There are still some confusing sentences. Incorrect comma usage is a common error, especially at the beginning of the manuscript. The patient enrollment numbers cited in the text on page 8 are confusing and do not agree with the numbers in Figure 2.

➔ As you pointed, there were some incorrect numbers on page 8. We corrected the figures, comma usage, and the confusing sentences.

Discretionary Revisions:
1. Figures 3 and 4 could be combined into parts A and B.
➔ We changed it to Fig 3a, b

Level of interest: An article of importance in its field
Quality of written English: Needs some language corrections before being published
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interests.
Reviewer’s report

Title: A Multi-center, Randomized, Clinical Study to Compare the Effect and Safety of Autologous Cultured Osteoblast Injection to Treat Fractures

Version: 5 Date: 7 October 2008

Reviewer: Peter Ebeling

Reviewer’s report:
The authors have answered all of my questions and the paper is much improved as a result.

Level of interest: An article of importance in its field

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests